Author's response to reviews

Title: Chronic virus infections suppress atopy but not asthma in a set of children from a large Latin American city

Authors:

Rafael V Veiga (rafael.veiga@uol.com.br)
Sérgio S Cunha (cunhatmt@hotmail.com)
Vitor CC Dattoli (vitor.dattoli@gmail.com)
Álvaro C Cruz (cruz.proar@gmail.com)
Philip J Cooper (pcooper@ecnet.ec)
Laura C Rodrigues (laura.rodrigues@lshtm.ac.uk)
Maurício L Barreto (mauricio@ufba.br)
Neuza M Alcantara-Neves (neuza@ufba.br)

Version: 6 Date: 11 March 2011

Author's response to reviews: see over
To Dr. Melissa Norton  
Editorial Manager (Medicine)  
Journal of Asthma  
BioMed Central

Dear Editor,

We are re-resubmitting the manuscript: 2056709653401617, entitled “Chronic virus infections supress atopy but not asthma in a set of children from a large Latin American city: A Cross-section study”, corrected according to the Editorial production team. Please see the corrections listed below.

Sincerely,

______________________________
Neuza M. Alcântara-Neves  
Instituto de Ciências da Saúde  
Universidade Federal da Bahia  
Avenida Reitor Miguel Calmon, sem n°, Canela  
CEP – 40110-100. Salvador, Bahia, Brazil  
E-mail: neuza@ufba.br
<table>
<thead>
<tr>
<th>Major revisions (we require the author to make these changes)</th>
<th>Answers</th>
</tr>
</thead>
<tbody>
<tr>
<td>Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution</td>
<td>This was done</td>
</tr>
<tr>
<td>Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)</td>
<td></td>
</tr>
<tr>
<td>Line numbers / track changes: please remove</td>
<td>This was done</td>
</tr>
<tr>
<td>Methodology: rename Methods</td>
<td>This was done</td>
</tr>
<tr>
<td>Highlighting: please remove</td>
<td></td>
</tr>
<tr>
<td>Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance</td>
<td>This was done</td>
</tr>
</tbody>
</table>